University of Toronto, Department of Electrical and Computer Engineering

ECE241F - Information Rele vant to Exam -CORRECTED

Fall 1999 A. Carusone, P. Lehn, J. Rose Exam Date and Time: Frida y December 17, 9:30am Sections of the T ext Covered & Labs

Chapter 1 Chapter 2 (except section 2.5.1) Chapter 3, Section 3.1-3.5, 3.6.2, 3.6.3, 3.6.4, 3.6.5, pages 100-106 of 3.8, 3.8.5, 3.8.8, pages 123-130 of 3.10 Chapter 4: 4.1, 4.2, 4.3, 4.4, 4.5, 4.6, 4.7.1 (including everything from 4.7 up to 4.7.1) Chapter 5 : 5.1 - 5.4 (except 5.1.3, 5.3.4) Chapter 6: 6.1 - 6.4 Chapter 7 : 7.1 - 7.9 (except 7.6) Chapter 8 : 8.1, 8.2, 8.4 (except 8.4.6, 8.4.7), 8.6.1 Chapter 10 : 10.1, 10.3.2, 10.3.4 You are also responsible for all of the material covered in labs 1 through 7.

Problems assigned bef ore the Midterm

Chapter 2: 2.2, 2.19, 2.21, 2.24, 2.25 Chapter 3: 3.1, 3.6, 3.8 and 3.12. Chapter 4: 4.2, 4.6, 4.11, 4.12, 4.18 Problems assigned since the Midterm

Chapter 5: 5,9,12,27 (CORRECTED) Chapter 7: 1,5, 15, 35, 36 Chapter 8: 1,5,14,15,29 (CORRECTED) Solutions to Assigned Pr oblems

Were/will be handed out in hard copy only, in class. They are not available online.